
Department Head:  Keith Gollwitzer
Machine Conditions: 12 Week Shutdown

○

• Shutdown Schedule
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• Feeder 24 - 7am
• F27 - not conventional power (lights)

� SEM hardware subtraction work in progress

○  Al 

� Extraction kicker upgrade decision made.  Will not do the upgrade this 
shutdown.   There have been a number of setbacks.   Will install the motorized 
stand for that kicker.   Also will open up magnet to take out RTV between the 
ferrites.

□ The present beam stop is an old SY sidewinder and is up to ATM.   
There are vacuum windows on either side.   WE are blowing up the 
beam in this location.   The upgrade will allow us to hold AP2 line 
vacuum.

� Status of the AP2 line beam stop.   Ready to go.   There are two of them 98% 
ready to go.    Doing some final adjustment on jaws, microswitches, etc.   The 

stands came in today.   Booster one and Pbar one are identical.  Two of the 
three have been built.  

○ Steve

� Calibrate loss monitors in the Accumulator.
� Fix broken loss monitors in the Debuncher.
� Need to check data took with IPM during the shutdown.

○ Vladimir

� Ps maintenance
� D:ELAM - rewired soft start so can turn it on quicker.
� Start on V901
� D:V714 and 716 - will fix the wiring.
� New DEX Bump PS will go in at AP10.

○ Bernie

Bill A.

• Progress
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New DEX Bump PS will go in at AP10.

� Will check d/a BPMs

○ Bill A.

� Repaired BAKER
�

○ Don Poll

� Replaced 20 BPMs
� Replacing boards
� Tor724 preAmps
� New amps to replace tronteks

○ Pat S.

� Emergency lights
� Java port
� Device database changes

○ Jim Budlong

� Ap50 SO cord - 25% complete
� AP10 dumb waiter fixed

○ Obie

� Trims in process of hanging the stands.   Hope to have one of them hung 
today.

○ Halbert

○

� Electronics for new Debuncher BPMs upstairs work in progress.

Dave P.

� Operatoins stuff

○ Stan Johnson

� Start by looking at EKIK

○ DVM

� Document running conditions - made the index.
� Documentation
� Run Co stuff

○ ME

○ Keith

○ Motorized kciker stand ekik
○ Rotate D/A BPMs
○ New D10 pipe - transverse schottky location
○ Quad bpm remove
○ Accumulator BPM maintenance
○ New trim PS
○ AP30 tank leak
○ 4-8GHz filter to get more gain
○ Replace IPM MCPs
○ LM calibration
○ AP2 Vacuum gate to help with furture removal of ic728
○ Ap2 trims
○ Install beam stop in ap2
○ Rotate ap2 bpm
○ OTR in AP1
○ Install new target
○ Ap50 SO cord
○ AP0 SO cord
○ Debuncher quad shunts
○ Feeder 24 replacement
○ LCW flush - DS end of AP2 especially
○ Survey level run
○ Install hall probes
○ Upgrade Deb. BPM
○ Beam line BPM cable

M:Q109 replacement

• Progress
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○ M:Q109 replacement
○ Cryo maintenace
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